Author's response to reviews

Title: Irresponsiveness of two retinoblastoma cases to conservative therapy correlates with up-regulation of hERG1 channels and of the VEGF-A pathway

Authors:

Pina Fortunato (pinfor@tin.it)
Serena Pillozzi (serena.pillozzi@unifi.it)
Angela Tamburini (a.tamburini@meyer.it)
Liliana Pollazzi (liliana.po@libero.it)
Agostino La Torre (agostino.latorre@unifi.it)
Alessandro Franchi (alessandro.franchi@unifi.it)
Annarosa Arcangeli (annarosa.arcangeli@unifi.it)

Version: 3 Date: 31 August 2010

Author's response to reviews: see over
Dear Editor,

We submit a revised version of our manuscript entitled “Irresponsiveness of two retinoblastoma cases to conservative therapy correlates with up-regulation of hERG1 channels and of the VEGF-A pathway”. We have addressed all the criticisms of the reviewers. In particular we have included comments both in the Introduction and in the Discussion as requested by Reviewer #1. Moreover the text and the figures have been revised and VEGF-A substituted to VEGF and arrows have been added in Figure 2 as requested by Reviewer #2.
Changes in the text are highlighted in red/bold.